Author's response to reviews

Title: A hidden HIV epidemic among women in Vietnam

Authors:

Thu Anh Nguyen (anhstat@yahoo.com)
Pauline Oosterhoff (pauline_oosterhoff@yahoo.com)
Anita Hardon (ahardon@xs4all.nl)
Hien Nguyen Tran (ngtrhien@yahoo.com)
Roel A Coutinho (roel.coutinho@rivm.nl)
Pamela Wright (pamela.wright@mcnv.nl)

Version: 4 Date: 17 January 2008

Author's response to reviews:

Dear Editors,

Thank you very much for your interest in publishing our paper A hidden HIV epidemic among women in Vietnam. Please see the revised manuscript for the changes made according to the instructions to the manuscript formatting.

We hope that these changes will be satisfactory to you.

With best regards,

Thu Anh Nguyen